Author’s response to reviews

Title: Comparison of the Glidescope(R) and Pentax AWS(R) laryngoscopes to the Macintosh laryngoscope for use by Advanced Paramedics in easy and simulated difficult intubation.

Authors:

Sajid Nasim (dr_snasim@yahoo.com)
Chrisen H Maharaj (chrisenm@gmail.com)
Muhammad A Malik (manwarmalik@yahoo.com)
John O'Donnell (john.odonnell@hse.ie)
Brendan D Higgins (brendan.higgins@nuigalway.ie)
John G Laffey (john.laffey@nuigalway.ie)

Version: 4 Date: 9 May 2009

Author's response to reviews:

Dear Editor,

Please find enclosed our revised manuscript entitled ‘Comparison of the Glidescope® and Pentax AWS® laryngoscopes to the Macintosh laryngoscope for use by Advanced Paramedics in easy and simulated difficult intubation’ by Nasim et al.

We have made the requested modifications to the manuscript as requested by the editorial team.

Best wishes,

John Laffey